Author's response to reviews

Title: Socioeconomic status and race/ethnicity independently predict health decline among older diabetics

Authors:

Emily J Nicklett (enicklet@umich.edu)

Version: 8 Date: 30 August 2011

Author's response to reviews:

The following text was added in the end of the Acknowledgements section:

This article was made available as Open Access with the support of the University of Michigan COPE Fund, http://lib.umich.edu/cope